Author’s response to reviews

Title: Developing a group intervention to manage fatigue in rheumatoid arthritis through modifying physical activity

Authors:

Victoria Salmon (v.salmon@exeter.ac.uk)
Sarah Hewlett (Sarah.hewlett@uwe.ac.uk)
Nicola Walsh (nicola.walsh@uwe.ac.uk)
John Kirwan (john.kirwan@bris.ac.uk)
Maria Morris (maria.morris@hotmail.co.uk)
Marie Urban (urban.zing.101@hotmail.co.uk)
Fiona Cramp (fiona.cramp@uwe.ac.uk)

Version: 3 Date: 28 Mar 2019

Author’s response to reviews:

Thank you for your comments and suggested revisions. I have made the following changes to the manuscript:

• I have added the following text to the start of the methods and acknowledgement sections respectively:

“Full methods are available elsewhere [21].”

“This research has been published in detail in VS’s PhD thesis [21].”

With regard to overlap with the PhD thesis we have attempted to minimise this where possible whilst maintaining a coherent manuscript. Further to this, the thesis does not include the detailed intervention mapping as presented in the paper and table 3.

• The document has been proofread and all mark up has been removed.

With Regards,

Victoria